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\Device Service Record Form for TB9NR002
	Date
	Comments

(Include the following in each DSR entry: What, Where, Why)
	Info Provided By

	5/5/2010
	Cavity moved from IB4 to ICB optical stand per Camille Ginsburg @ 1530 hrs
	RE / LZ

	5/14/2010
	TB9NR001 and TB9NR002 QC discussed in meeting with vendor 

summary slides compiled by Leonardo Ristori
	Camille Ginsburg

	6/17/10
	Cavity moved from ICB to Shipping to be sent to JLAB. MMR 147606
	L. Zimmerman

	6/23/2010
	Rongli Geng confirmed receipt of TB9NR002 at JLab, where it will receive a 10 micron internal BCP.  The cavity is currently as-received from Niowave-Roark.
	Camille Ginsburg

	6/30/2010
	Since it was shipped to JLab, the status is now “reserved: offsite”.
	Jamie Blowers

	7/15/10
	Cavity received 10 micron BCP @ JLAB, and has been returned. Currently residing on ICB production floor
	Damon Bice

	7/15/2010
	Cavity has been moved to the Optical Inspection Stand per Camille Ginsburg @ 1410 hours
	Rick Espinoza

	7/19/2010
	Previously observed irregularities had been re-examined. Most of them got removed by BCP “wash” at JLAB. Signs of welding overlap instabilities in the end-cells are still there (as expected for only 10um BCP). Cavity is ready for pick-up off optical inspection bench.
	Dmitri A. Sergatskov

	7/20/2010
	Dmitri and I crated the cavity in an instrument crate.  The flanges have plastic covers.  The cavity volume is at atmospheric pressure.  It is ready to be moved to ANL for bulk-EP.
	Camille Ginsburg

	7/21/10
	Cavity repacked into wooden crate and transported to ANL for bulk EP / HPR
	Damon Bice

	7/27/2010
	Tom Reid spreadsheet shows 120 um bulk EP on 7/27/2010
	Camille Ginsburg

	7/29/10
	Cavity received ~120 microns EP, 2 pass HPR, transported to ICB awaiting JLAB furnace treatment
	Damon Bice

	8/5/10
	Jim Rife performed an optical inspection.  
	Camille Ginsburg

	8/6/10
	Cavity moved from ICB to shipping to be sent to JLAB. MMR 147627
	L. Zimmerman

	8/12/10
	Rongli confirmed the cavity has arrived at JLab.
	Camille Ginsburg

	9/28/2010
	Lucas Zimmerman confirmed the cavity is back at Fermilab, in ICB.
	Camille Ginsburg

	Oct-06-2010
	On the optical inspection stand (ICB).
	Dmitri Sergatskov

	10/18/10
	Transported to ANL to receive 20 micron EP, HPR, and assembly 
	Damon Bice

	10/27/10
	Cavity received 20 um EP, HPR, assy sequence@ ANL. Transported to IB1 for 120c bake and VTS
	Damon Bice

	10/27/10
	Cavity received, prepping for 120 bake 
	M carter

	11/1/10
	VTS e-log entry by J. Ozelis:

In the pi-mode, the cavity was limited to 29.9 MV/m by quench - possibly FE-induced. There was a brief burst of FE at 25MV/m, which soon processed away, but FE returned thereafter, but never exceeded 10mR/hr. OST's indicated cell #8 as the possible quench location, about 180 degrees from the FPC centerline. Low field Q was ~2E10, Q at quench was ~ 1.2E10.  Mode measurements showed all cell pairs reaching 33-37 MV/m. The fact that the pi-mode limit (30 MV/m) is lower than the mode-limit of cell #8 (32.8 MV/m) may be due to the influence of  FE which was stronger in the pi-mode than in other modes (basically non-existent except in the 2/9 mode).

Ozelis slides
	Camille Ginsburg

	11/15/10
	Cavity is let up, both end flanges removed for op insp, ready for pickup
	M carter

	11/19/10
	Cavity picked up for transport to ICB
	M carter

	11/19/10
	Cavity moved from IB1 to ICB Optical Inspection Area
	L. Zimmerman

	11/24/10
	Inspection completed. Cavity is at the inspection system and can be moved anytime
	E. Toropov

	12/02/10
	Cavity is in the box near the optical inspection system
	E. Toropov

	4/20/11
	Cavity transported from ICB to ANL for 30um EP / HPR / Assy sequence
	Damon Bice

	4/21/11
	Cavity received 30um EP, HPR, Assembly at ANL
	Chris Baker

	4/29/2011
	Cavity transported from ANL to ICB per Damon Bice @ 1030 hours
	Rick Espinoza

	5/11/11
	Cavity at ICB.  Scheduled to be moved to IB1 12-May-2011 for 120C bake.  VT scheduled 20-May-2011.  See Ginsburg Cavity Scheduling Mtg. 4-May-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 4-May-2011.

	5/17/2011
	Cavity has been moved from ICB to IB1 per Dan Marks @ 1315 hours
	Rick / Luke

	5/17/2011
	Cavity received at IB1
	D. Marks

	
	
	

	5/23/11
	Cavity at IB1 for 120C bake then VT.  VT scheduled 24-May-2011.  See Ginsburg Cavity Scheduling Mtg. 18-May-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 18-May-2011.

	5/25/11
	Cavity will be sent back to ANL due to a potential leak. The cavity is now crated and awaiting pick up from IB1.
	D. Marks



	5/26/11
	Cavity moved from IB1 to ICB per Daniel Marks
	J.Brown

	6/9/11
	Cavity in ICB.  Return to ANL to repair vacuum leak.  See Ginsburg Cavity Scheduling Mtg. 1-June-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 1-June-2011.

	6/15/11
	Cavity in ICB.  Return to ANL to repair vacuum leak.  See Ginsburg Cavity Scheduling Mtg. 15-June-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 15-June-2011.

	6/29/11
	Cavity in ICB.  Return to ANL to repair vacuum leak.  See Ginsburg Cavity Scheduling Mtg. 29-June-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 29-June-2011.

	7/13/11
	Cavity in ICB.  Return to ANL to repair vacuum leak.  See Ginsburg Cavity Scheduling Mtg. 13-July-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 13-July-2011.

	8/24/2011
	Cavity transported from ICB to ANL per Damon Bice @ 1440 hours
	Rick Espinoza

	8/30/11
	Cavity at ANL.  Next: VT prep (single pass HPR), VT.  See Ginsburg Cavity Scheduling Mtg. 24-Aug-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 24-Aug-2011

	8/30/11
	Cavity had bottom B.L. flanges assembly removed, 2 passy HPR, VTS assy
	Damon Bice

	8/31/2011
	Cavity transported from ANL to IB1 per Damon Bice @ @ 1130 hours
	Rick Espinoza

	9/6/2011
	Cavity was quench limited to 24MV/m, Field emission began at 18MV/m, 

and was intense enough (20mR/hr) to lead to several FE-induced quenches below 20MV/m. Also saw some MP from 21-22 MV/m. Pulse-processing of FE successfully reduced it, and quench limit at 24MV/m was accompanied by only minimal radiation. Second sound suggest cell #6 as quench origin.  Low field Q was ~2E10; Q at quench was ~1.2E10. 
Ozelis test report
	J. Ozelis

	9/7/11
	VT 6.Sept.2011.  Quench limited to 24 MV/m; cell 6 limitation from second sound (different from previous test); multipacting also seen 20-22 MV/m range with second sound processed away; FE processed mostly away.  Next: optical inspection.  See Ginsburg Cavity Scheduling Mtg. 7-Sept-2011 slides for details.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 7-Sept-2011

	9/13/11
	Cavity moved from IB1 to ICB
	LZ / RE

	9/21/11
	Cavity moved to Optical Inspection Station 9/21/11.
	Jim Rife

	2011-09-22
	Inspection is performed. A bunch of strange-looking features are on the surface:

( Q:\TD_SCRF\Cavity Imaging\Fermilab\9-cell cavity\tb9nr002\tb9nr002_2011-09-21 )

Similar black dots could appear on other cavities but they are usually much less in size. They look like bumps, not pits to me.
	E Toropov

	10/10/11
	Cavity at ICB.  Optical inspection done : bunch of strange looking surface features (large-ish black dots) seen, guess bumps rather than pits.  TB9NR002, cell 6 @ ~60 deg: Next: depending on TB9NR001 perhaps tumbling.  See Ginsburg Cavity Scheduling Mtg. 5-Oct-2011 slides for details.
	Jim Rife

	12/19/11
	Next: depending on TB9NR001 perhaps tumbling.
	Jim Rife per Ginsburg Cavity Scheduling Mtg. 14-Dec-2011

	1/27/12
	Moved cavity to IB4 CPL tumbling area.
	G.Steuer

	2-29-12 
	Prepped  cavity for tumbling 1st step ACT M tumbled for 7 hours, cleaned 
	Dburk

	3-13-12
	Prepped cavity for step 2 cones and tumbled for 20 hours cleaned in the vertical ultrasonic cleaner, left in tumbling frame and placed in crate
	Dburk

	8/16/12
	US cleaned for 1 hour 
	Chad Thompson

	8/17/12
	US cleaned for 1 hour
	Chad Thompson

	8/20/12
	US cleaned for 1 hour there was still media in the HOM cans used Q-tip to remove this cleaning removed leftover media
	Dburk

	8/21/12
	Cavity location verification.  Cavity @ IB.
	Jim Rife

	8-27-12
	Removed 4-bar frame and placed into crate. Moved to ICB OPT Insp.
	Dburk

	8/27/12
	Moved to ICB for Optical inspection.  Returned to CPRL.
	Chad Thompson

	8/29/12
	Loaded with 800 mesh Al and wood blocks, tumbled for 35 hours.
	Chad Thompson

	8/31/12
	US cleaned for two hours.  Left in US cleaner awaiting HPR.
	

	9/4/12
	Cavity was moved to cleanroom for HPR and allowed to dry.
	Chad Thompson

	9/4/12
	Moved to ICB for Optical inspection.  
	Chad Thompson

	9/5/12
	Moved to CPRL.  Loaded with col silica and wood blocks, started tumbling
	Chad Thompson

	9/8/12
	Tumbling stopped at 70hr due to broken gear.  Cavity removed and low pressure rinsed.  Cavity stored in water in US cleaner.
	Chad Thompson

	9/11/12
	Cavity moved to cleanroom for 2 pass HPR
	Chad Thompson

	9-12-12
	Cavity left to dry in clean room
	Dburk

	9/27/12
	Cavity moved to CPRL tumbling area in preparation for tumbling
	Chad Thompson

	10/29/12
	Prepped cavity for step 1 - Act M and tumbled for 8 hours, cleaned in the vertical ultrasonic cleaner for 1 hour, left in tumbling frame
	Chad Thompson

	10/30/12
	Prepped cavity for step 2 - RG 22 ½ cones and tumbled for 12 hours, cleaned in the vertical ultrasonic cleaner for 1 hour, changed the water in the US cleaner, left in tumbling frame
	Chad Thompson

	10/31/12
	Prepped cavity for step 3 - 800 mesh and wood blocks and tumbled for 30 hours, cleaned in the vertical ultrasonic cleaner for 2 hours, changed water and US cleaned for another 2 hours, left in tumbling frame
	Chad Thompson

	11/06/12
	Prepped cavity for step 4 – colloidal silica and wood blocks and tumbled for 71 hours
	Chad Thompson

	11/09/12
	Cleaned in the vertical ultrasonic cleaner for 1 hour, left hanging in tumbling frame in the US cleaner so HPR can be started on Monday.
	Chad Thompson

	11/12/12
	Moved cavity to IB4 CPL cleanroom and began first pass w/HPR. Will continue with final 2 passes tomorrow morning.
	G.Steuer

	11/14/12
	Completed final 2 passes on HPR and let cavity dry overnight.
	G.Steuer

	11/15/12
	Moved Cavity to ICB Optical Inspection lab and began inspection.
	G.Steuer

	11/16/12
	Moved cavity back to IB4 CPL tumbling area after Opt.Insp. was completed
	G.Steuer

	11/19/12
	Moved cavity to ICB for Optical Inspection per C.Cooper request.
	G.Steuer

	11/21/12
	Completed addition images of equator and iris and moved back to IB4 CPL tumbling area.
	G.Steuer

	12/5/12
	Moved cavity to ICB storage rack per C.Cooper’s request.
	G.Steuer
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